Author's response to reviews

Title: Mosquito larvicidal and antimicrobial activity of protein of Solanum villosum leaves

Authors:

Nandita NC Chowdhury (cnandita_2006@yahoo.co.in)
Subrata SL Laskar (slaskar01@yahoo.co.in)
Goutam GC Chandra (goutamchandra63@yahoo.co.in)

Version: 4 Date: 29 November 2008

Author's response to reviews:

To
The Editor, 29.11.2008
BMC Complementary and Alternative Medicine
Respected Sir/Madam,

Sending herewith the revised and fully formatted article entitled “Mosquito larvicidal and antimicrobial activity of protein of Solanum villosum leaves”, for publication in your esteemed journal. The paper has been revised carefully and all the formatting changes have been made.

Please acknowledge the receipt of this mail.

Looking forward to hearing from you,

With regards.

Yours faithfully

Goutam Chandra, Ph.D.
Mosquito and Microbiology Research Units,
Parasitology Laboratory,
Department of Zoology,
The University of Burdwan,
West Bengal, India
Email: goutamchandra63@yahoo.co.in